Author’s response to reviews

Title: The prognostic significance of the neutrophil-to-lymphocyte ratio and the platelet to-lymphocyte ratio in giant cell tumor of the extremities

Authors:
Zhenhao Chen (chenzhenhao17@126.com)
Guanglei Zhao (gzhao@fudan.edu.cn)
Jun Xia (hudbt17089@gmail.com)
Feiyan Chen (chenfy73@163.com)
Li Jiang (charleyii@163.com)

Version: 2 Date: 02 Mar 2019

Author’s response to reviews:

Thank you again for the review. We have made corrections according to your suggestion as follows,

Editor Comments: Unfortunately, there are still some minor language issues throughout the manuscript. Please ensure that you have thoroughly checked your manuscript for any other language errors. We recommend that you ask a native English speaking colleague to help you copy edit the paper. If this is not possible, you may need to use a professional language editing service. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication.

Response: Many thanks for your time and consideration. I hope that we have corrected the errors in the article this time.

Reviewer reports:

Fei Gao (Reviewer 1): There are still so many English errors including words and grammars in this paper that must be corrected and I strongly recommend it should be reviewed by a native English speaker or professionals to further modify the manuscript.

Response: Sorry for the English errors we had made. It has been reviewed by professionals now. I hope that we have corrected the errors in the article this time. Many thanks for your criticism and suggestions.
Dedong Cao, M.D. (Reviewer 2): The concerns have been answered properly.

Response: Thank you for affirming our work. Your questions and suggestions helped us improve the quality of our article.